\begin{thebibliography}{10}
\expandafter\ifx\csname url\endcsname\relax
  \def\url#1{\texttt{#1}}\fi
\expandafter\ifx\csname urlprefix\endcsname\relax\def\urlprefix{URL }\fi

\bibitem{Schmidt:01m}
D.~C. Schmidt, R.~Schantz, M.~Masters, J.~Cross, D.~Sharp, L.~DiPalma, {Towards
  Adaptive and Reflective Middleware for Network-Centric Combat Systems},
  CrossTalk.

\bibitem{Gill:01a}
J.~Loyall, J.~Gossett, C.~Gill, R.~Schantz, J.~Zinky, P.~Pal, R.~Shapiro,
  C.~Rodrigues, M.~Atighetchi, D.~Karr, {Comparing and Contrasting Adaptive
  Middleware Support in Wide-Area and Embedded Distributed Object
  Applications}, in: Proceedings of the 21st International Conference on
  Distributed Computing Systems (ICDCS-21), IEEE, 2001, pp. 625--634.

\bibitem{Schantz:04}
P.~Sharma, J.~Loyall, G.~Heineman, R.~Schantz, R.~Shapiro, G.~Duzan,
  {Component-Based Dynamic QoS Adaptations in Distributed Real-Time and
  Embedded Systems}, in: {Proc. of the Intl. Symp. on Dist. Objects and
  Applications (DOA'04)}, Agia Napa, Cyprus, 2004.

\bibitem{ogata97}
K.~Ogata, Modern Control Engineering, Prentice Hall, Englewood Cliffs, NJ,
  1997.

\bibitem{Li:99}
B.~Li, K.~Nahrstedt, {A Control-based Middleware Framework for QoS
  Adaptations}, IEEE Journal on Selected Areas in Communications 17~(9) (1999)
  1632--1650.

\bibitem{Tarek:03}
T.~Abdelzaher, K.~Shin, N.~Bhatti, {User-Level QoS-Adaptive Resource Management
  in Server End-Systems}, {IEEE Transactions on Computers} 52~(5).

\bibitem{Sha:94}
L.~Sha, R.~Rajkumar, S.~S. Sathaye, {Generalized Rate Monotonic Scheduling
  Theory: A Framework for Developing Real-time Systems}, {Proceedings of the
  IEEE} 82~(1).

\bibitem{Schechter:98}
S.~S, M.~Krishnan, M.~D. Smith, {Using path profiles to predict http requests},
  in: Proceedings of the Seventh International World Wide Web Conference (WWW
  98), Brisbane, Australia, 1998.

\bibitem{desidirata:98}
L.~Welch, B.~Shirazi, B.~Ravindran, C.~Bruggeman, {DeSiDeRaTa: QoS Management
  Technology for Dynamic, Scalable, Dependable, Real-Time Systems}, in:
  {Proceedings of the 15th IFAC Workshop on Distributed Computer Control
  Systems (DCCS'98)}, 1998.

\bibitem{Nutt:00}
G.~Nutt, S.~Brandt, A.~Griff, S.~Siewert, T.~Berk, M.~Humphrey, {Dynamically
  Negotiated Resource Management for Data Intensive Application Suites}, in:
  {IEEE Transactions on Knowledge and Data Engineering}, 2000.

\bibitem{Hansen:01}
J.~Hansen, J.~Lehoczky, R.~Rajkumar, {Optimization of Quality of Service in
  Dynamic Systems}, in: {Proceedings of the 9th International Workshop on
  Parallel and Distributed Real-Time Systems}, 2001.

\bibitem{Ghosh:04}
S.~Ghosh, R.~Rajkumar, J.~Hansen, J.~Lehoczky, {Adaptive QoS Optimizations for
  Radar Tracking}, in: {Proceedings of the 10 International Conference on
  Real-time and Embedded Computing Systems (RTCSA'04)}, 2004.

\bibitem{Schmidt:03m}
N.~Wang, D.~C. Schmidt, A.~Gokhale, C.~Rodrigues, B.~Natarajan, J.~P. Loyall,
  R.~E. Schantz, C.~D. Gill, {QoS-enabled Middleware}, in: Q.~Mahmoud (Ed.),
  Middleware for Communications, Wiley and Sons, New York, 2003, pp. 131--162.

\bibitem{Schmidt:05y}
G.~Deng, J.~Balasubramanian, W.~Otte, D.~C. Schmidt, A.~Gokhale, {DAnCE: A
  QoS-enabled Component Deployment and Conguration Engine}, in: Proceedings of
  the 3rd Working Conference on Component Deployment, Grenoble, France, 2005.

\bibitem{Wang:03b}
N.~Wang, C.~Gill, {Improving Real-Time System Configuration via a QoS-aware
  CORBA Component Model}, in: {Hawaii International Conference on System
  Sciences, Software Technology Track, Distributed Object and Component-based
  Software Systems Minitrack, HICSS 2003}, HICSS, Honolulu, HW, 2003.

\bibitem{CCM-light:03}
{Object Management Group}, {Light Weight CORBA Component Model Revised
  Submission}, {OMG Document realtime/03-05-05} Edition (May 2003).

\bibitem{RTCorba:02}
{Object Management Group}, {Real-time CORBA Specification}, {OMG Document
  formal/02-08-02} Edition (Aug. 2002).

\bibitem{Markus:02}
M.~Volter, A.~Schmid, E.~Wolff, {Server Component Patterns: Component
  Infrastructures Illustrated with EJB}, Wiley Series in Software Design
  Patterns, West Sussex, England, 2002.

\bibitem{Schmidt:00a}
D.~C. Schmidt, M.~Stal, H.~Rohnert, F.~Buschmann, {Pattern-Oriented Software
  Architecture: Patterns for Concurrent and Networked Objects, Volume 2}, Wiley
  \& Sons, New York, 2000.

\bibitem{Schmidt:04w}
N.~Wang, C.~Gill, D.~C. Schmidt, V.~Subramonian, {Configuring Real-time Aspects
  in Component Middleware}, in: {Proc. of the International Symposium on
  Distributed Objects and Applications (DOA'04)}, Agia Napa, Cyprus, 2004.

\bibitem{DandC:03}
{Object Management Group}, {Deployment and Configuration Adopted Submission},
  {OMG Document ptc/03-07-08} Edition (Jul. 2003).

\bibitem{Liu:00}
J.~W.~S. Liu, Real-Time Systems, Prentice Hall, New Jersey, 2000.

\bibitem{Das:2005}
B.~Dasarathy, S.~Gadgil, R.~Vaidhyanathan, K.~Parmeswaran, B.~Coan, M.~Conarty,
  V.~Bhanot, {Network QoS Assurance in a Multi-Layer Adaptive Resource
  Management Scheme for Mission-Critical Applications using the CORBA
  Middleware Framework}, in: {Proceedings of the IEEE Real-Time Technology and
  Applications Symposium (RTAS)}, IEEE, San Francisco, CA, 2005.

\bibitem{JunSun:97}
J.~Sun, Fixed-priority end-to-end scheduling in distribured real-time systems,
  Ph.D. thesis, Department of Computer Science, University of Illinois at
  Urbana-Champagne (1997).

\bibitem{Schantz:98}
R.~Vanegas, J.~Zinky, J.~Loyall, D.~Karr, R.~Schantz, D.~Bakken, {Quo's Runtime
  Support for Quality of Service in Distributed Objects}, in: {Proceedings of
  the IFIP International Conference on Distributed Systems Platforms and Open
  Distributed Processing (Middleware'98)}, IFIP, The Lake District, England,
  1998.

\bibitem{Paulo:03}
P.~Stadzisz, J.~Simao, M.~Quinaia, {A Pattern System to Supervisory Control of
  Automated Manufacturing System}, in: {Proceedings of the Third Latin American
  Conference on Pattern Languages of Programming}, Porto de Galinhas, Brazil,
  2003.

\bibitem{Cross:01}
{Joseph K. Cross and Patrick Lardieri}, {Proactive and Reactive Resource
  Reallocation in DoD DRE Systems}, in: Proceedings of the OOPSLA 2001 Workshop
  "Towards Patterns and Pattern Languages for OO Distributed Real-time and
  Embedded Systems", 2001.

\bibitem{Zinky:2002}
J.~Zinky, J.~Loyall, R.~Shapiro, {Runtime Performance Modeling and Measurement
  of Adaptive Distributed Object Applications}, in: {Proceedings of the
  International Symposium on Distributed Objects and Applications (DOA'2002)},
  Irvine, CA, 2002.

\bibitem{Khanna:92}
{Khanna, S., {\em et al.}}, {Realtime Scheduling in SunOS 5.0}, in: Proceedings
  of the USENIX Winter Conference, USENIX Association, 1992, pp. 375--390.

\bibitem{Kenyon:96}
C.~Kenyon, {Best-fit bin-packing with random order}, in: {Proceedings of the
  Seventh Annual ACM-SIAM Symposium on Discrete Algorithms}, 1996.

\bibitem{Baker:85}
B.~S. Baker, {A New Proof for the First-Fit Decreasing Bin-Packing Algorithm},
  in: {Journal of Algorithms}, 1985.

\bibitem{moreira97dynamic}
J.~E. Moreira, V.~K. Naik, Dynamic resource management on distributed systems
  using reconfigurable applications, IBM Journal of Research and Development
  41~(3) (1997) 303--330.
\newline\urlprefix\url{citeseer.ist.psu.edu/moreira97dynamic.html}

\bibitem{Banga:99}
G.~Banga, P.~Druschel, J.~Mogul, {Resource Containers: A New Facility for
  Resource management in Server Systems}, in: Proceedings of the Symposium on
  Operating Systems Design and Implementation (OSDI 99), 1999.

\bibitem{Welch:98}
L.~R. Welch, B.~A. Shirazi, B.~Ravindran, C.~Bruggeman, {DeSiDeRaTa: QoS
  Management Technology for Dynamic, Scalable, Dependable Real-Time Systems},
  in: {IFACs 15th Symposium on Distributed Computer Control Systems (DCCS98)},
  IFAC, 1998.

\bibitem{kyhcm:01:coots}
F.~Kon, T.~Yamane, C.~Hess, R.~Campbell, M.~D. Mickunas, {Dynamic Resource
  Management and Automatic Configuration of Distributed Component Systems}, in:
  Proceedings of the 6th USENIX Conference on Object-Oriented Technologies and
  Systems (COOTS'2001), San Antonio, Texas, 2001, pp. 15--30.
\newline\urlprefix\url{citeseer.ist.psu.edu/341630.html}

\bibitem{shirazi99dynbench}
B.~Shirazi, L.~R. Welch, B.~Ravindran, C.~Cavanaugh, B.~Yanamula, R.~Brucks,
  E.~nam Huh, Dynbench: A dynamic benchmark suite for distributed real-time
  systems, in: {IPPS}/{SPDP} Workshops, 1999, pp. 1335--1349.
\newline\urlprefix\url{citeseer.ist.psu.edu/article/shirazi99dynbench.html}

\bibitem{Schmidt:04z}
K.~Balasubramanian, J.~Balasubramanian, J.~Parsons, A.~Gokhale, D.~C. Schmidt,
  {A Platform-Independent Component Modeling Language for Distributed Real-time
  and Embedded Systems}, in: Proceedings of the 11th Real-time Technology and
  Application Symposium (RTAS '05), IEEE, San Francisco, CA, 2005.

\bibitem{Schmidt:05u}
N.~Shankar, J.~Balasubramanian, D.~C. Schmidt, G.~Biswas, P.~Lardieri,
  E.~Mulholland, T.~Damiano, {A Framework for (Re)Deploying Components in
  Distributed Real-time and Embedded Systems}, in: Poster paper in the
  Dependable and Adaptive Distributed Systems Track of the 21st ACM Symposium
  on Applied Computing, Dijon, France, 2005.

\bibitem{Hill:05}
J.~Hill, J.~Slaby, S.~Baker, D.~Schmidt, {Applying System Execution Modeling
  Tools to Evaluate Enterprise Distributed Real-time and Embedded Systems QoS},
  Tech. Rep. ISIS-05-604, Vanderbilt University (2005).

\end{thebibliography}

% $Id: intro.tex,v 1.27 2005/11/23 19:54:35 gokhale Exp $

\section{Introduction}
\label{sec:intro}

\subsection{Characteristics and Challenges of Enterprise DRE Systems}
\label{sec:trends}

Enterprise distributed real-time and embedded (DRE) systems, such as
shipboard computing environments~\cite{Schmidt:01m}, airborne command
and control systems~\cite{Gill:01a}, and intelligence, surveillance
and reconnaissance systems~\cite{Schantz:04}, are growing in
complexity and importance as more computing devices are networked
together to help automate tasks previously done by human operators.
These types of systems are characterized by stringent
quality-of-service (QoS) requirements, such as low latency and jitter,
expected in real-time and embedded systems, as well as high
throughput, scalability, and reliability expected in enterprise
distributed systems.  In particular, these types of systems share the
following characteristics:

% changed to cramplist everywhere since it was causing things to be
% too near to each other
\begin{itemize}\cramplist
\item {\bf Heterogeneity.} Enterprise DRE systems often run on a
variety of computing platforms that are interconnected by different
types of networking technologies with varying quality of service (QoS)
properties.  The efficiency and predictability of enterprise DRE
systems built using different infrastructure components varies
according to the type of computing platform and interconnection
technology.

\item {\bf Networked embedded properties.}  Enterprise DRE systems are
often composed of multiple networked embedded subsystems.  For
example, a shipboard computing environment may have multiple {\em
threat tracking subsystems} that colloborate with multiple {\em
mission planner subsystems} (which make decisions on what actions to
take depending on the threats being tracked) and multiple {\em effector
subsystems} (which enact the decisions).  There should be strong
communication and colloboration between these networked embedded
subsystems to achieve the critical functionalities of an enterprise
DRE system.

\item {\bf Simultaneous support for multiple QoS properties.}
Software controllers~\cite{ogata97} are increasingly replacing
mechanical and human control of enterprise DRE systems.  These
controllers must simultaneously support many challenging QoS
constraints, including (1) {\em real-time requirements}, such as low
latency and bounded jitter, (2) {\em availability requirements}, such
as fault propagation/recovery across distribution boundaries, (3) {\em
security requirements}, such as appropriate authentication and
authorization, and (4) {\em physical requirements}, such as limited
weight, power consumption, and memory footprint.  For example, a
distributed patient monitoring system requires predictable, reliable,
and secure monitoring of personal health data that can be distributed
in a timely manner to healthcare providers.

\item {\bf Large-scale system-of-systems operation.}  The scale and
complexity of enterprise DRE systems often makes it infeasible to
deploy them in separate disconnected islands.  The functionality of
these systems must therefore be partitioned and distributed over a
range of networks.  For example, an urban bio-terrorist evacuation
capability requires highly distributed functionality involving
networks connecting command and control centers with bio-sensors that
collect data from police, hospitals, and urban traffic management
systems.

\item {\bf Dynamic operating conditions.}  Operating conditions for
enterprise DRE systems can change dynamically, resulting in the need
for appropriate adaptation and resource management strategies for
continued successful system operation.  In civilian contexts, for
instance, power outages underscore the need to detect failures in a
timely manner and adapt in real-time to maintain mission-critical
power grid operations.  In military contexts, likewise, a mission mode
change or loss of functionality due to an attack in combat operations
requires adaptation and possible resource reallocation to continue
with mission-critical capabilities.
\end{itemize}

As evidenced from the discussion above, enterprise DRE systems have a
range of QoS requirements that may vary at runtime due to
planned~\cite{Li:99} and unplanned~\cite{Tarek:03} events.  Examples
of planned events include mission goal changes due to refined
intelligence and planned task-completion exceeding mission parameters.
Likewise, examples of unplanned events might include system runtime
performance changes due to loss of resources, transient overload,
and/\-or changes in algorithmic parameters (such as modifying an air
threat tracking subsystem to have better coverage).

Due to the challenges of applying theoretically-grounded techniques
for generating and adapting resource allocations to enterprise DRE
systems, conventional development and quality assurance processes
employ {\em ad hoc} techniques comprising heuristics and/\-or simple
planning algorithms to solve specific resource management problems
identified at design-time.  Unfortunately, these techniques are poorly
suited to handle dynamic events (such as equipment failure, mission
changes, damage, or significant overload) that enterprise DRE systems
encounter at runtime.

\subsection{Addressing DRE System Challenges with Dynamic Resource Management}
\label{sec:motiv-chal}

Approaches to resolving the challenges outlined in
Section~\ref{sec:trends} have traditionally focused on conducting
offline static resource-management analysis to determine how the
system should respond to planned events and a limited set of unplanned
events~\cite{Sha:94}.  Such analysis is performed by complex
analytical and simulation models of the systems, subjecting these
models to workloads that represent various planned and unplanned
events and determining how well the systems perform in the face of the
emulated mission-level and environmental changes.  As the size of the
enterprise DRE systems and the environments in which they execute
increases, however, offline static analysis of the system becomes
intractable due to the explosion of possible state permutations and
the limitations of analytical methods to scale seamlessly to cover
large state-spaces~\cite{Schechter:98}.

{\em Dynamic resource management}
(DRM)~\cite{desidirata:98,Nutt:00,Hansen:01} is a promising paradigm
for alleviating the limitations of static analysis and support
different types of applications running in enterprise DRE system
environments - as well as to optimize and reconfigure the resources
available in the system to meet the changing needs of applications at
runtime.  A key goal of DRM is to ensure that enterprise DRE systems
can adapt dependably in response to dynamically changing conditions
({\em e.g., evolving mutli-mission priorities}) to ensure that
computing and networking resources are best aligned to meet critical
mission requirements.  A key assumption in DRM technologies is that
the levels of service in one dimension can be coordinated with and/or
traded off against the levels of service in other dimensions to meet
mission needs, {\em e.g.}, the security and dependability of message
transmission may need to be traded off against latency and
predictability.

\iffalse 
% 
Capturing explicit descriptions system behavioral and performance
characteristics (e.g. each software component's resource usage and
performance profile) in a computable representation

- how components consume resources in response different stimulation
  (e.g., method invocations or event reception).

- how to designate and quantify critical system flows (i.e., the
  problem CUTS addresses)

- how to capture the tasking and concurrency architecture (a.k.a., the
  system execution model) to enable validation.

- how to manage the evolution of all this data over the design
  lifecycle: starting with speculation, going to benchmarking, actual
  components etc.

Validating the dynamically generated resource allocations do not
violate critical system properties (e.g., meeting real-time deadlines)

- how to move work traditionally done by human designers into run time
  services

- how to develop computationally efficient mathematical models that
  can accurately predict run-time system properties of a given
  resource allocation

- how to recover when validation fails

Identifying the abstract base capabilities that enable dynamic
resource management services to be tailored to different system
requirements via configuration (as opposed to redevelopment).

- what are the right points of variability and what abstractions best
  capture that variability.

- devising information models that suit the human specification and
  the run time services

- defining a development work-flow that orchestrates the activities of
  the system designer, the DRM service software developer, and
  operation of the run-time DRM services

Mapping abstract allocations into various application middleware,
operating systems and networking environments.
\fi

Prior work~\cite{desidirata:98,Nutt:00,Hansen:01} on DRM has focused
on general techniques for QoS-enabled enterprise DRE systems or
specialized techniques for niche real-time systems~\cite{Ghosh:04}.
Our work in this paper augments this prior work by focusing on (1)
providing a standards-based {\em multi-layer resource management}
(MLRM) infrastructure for configurable DRM services and (2) developing
and empirically validating theoretically-grounded technologies for
applying these DRM services to mission-critical enterprise DRE
systems.  In particular, our approach to DRM described in this paper
innovates across the following interrelated technology areas:
\begin{itemize}\cramplist
\item {\bf Task allocation and analysis techniques} that generate and
   validate a coordinated set of resource allocations to satisfy
   evolving mission functional and performance requirements.

\item {\bf QoS-enabled component middleware} that provides a
   configurable computing infrastructure capable of enforcing resource
   allocations, even in the presence of overload and failures, thereby
   enabling and sustaining effective enterprise DRE software
   operation.

\iffalse
\item {\bf Model-based computing tools} to provide an environment for
   tailoring application designs toward patterns and workflows that
   enable fine-grained dynamic resource management; and to enable the
   specification of application metadata to describe operational modes
   and performance requirements for both applications and systems.
\fi
\end{itemize}

The remainder of this paper describes how our MLRM architecture has
demonstrated DRM capabilities in a {\em total ship computing
environment} (TSCE), which is a grid of computers that manage many
aspects of a ship's power, navigation, command and control, and
tactical operations~\cite{Schmidt:01m}.  To make a TSCE an effective
platform requires coordinated and standards-based DRM services that
can support multiple QoS requirements, such as survivability,
predictability, security, and efficient resource utilization.  The
MLRM described in this paper was developed for the DARPA's {\em
Adaptive and Reflective Middleware Systems} (ARMS) program
(\url{dtsn.darpa.mil/ixodarpatech/ixo_FeatureDetail.asp?id=6}), which
is applying DRM technologies to coordinate a grid of computers that
manage and automate many aspects of Naval TSCEs.  We describe and
empirically evaluate how the ARMS MLRM manages the computing resources
in TSCE dynamically and ensures proper execution of TSCE missions in
response to mission mode changes and/\-or resource load changes and
failures, as well as capability upgrades.

\iffalse
% Guys, I've commented this out to save space.
\subsection{Paper Organization}
\label{sec:org}

The remainder of the paper is organized as follows:
Section~\ref{sec:mlrm} describes the design goals and components of
the ARMS MLRM architecture in more detail and shows how it has been
applied to a total ship computing environment; Section~\ref{sec:eval}
empirically evaluates the ARMS MLRM architecture in a set of
representative experiments; Section~\ref{sec:relwork} compares our
work with related research; and Section~\ref{sec:conclu} presents
concluding remarks.
\fi

%%% Local Variables: 
%%% mode: latex
%%% TeX-master: "JSS-2006"
%%% End: 
